Author's response to reviews

Title: COMMAND: a multi-centre, randomised controlled trial of cognitive therapy to prevent harmful compliance with command hallucinations

Authors:

Max Birchwood (m.j.birchwood.20@bham.ac.uk)
Emmanuelle Peters (emmanuelle.peters@kcl.ac.uk)
Nicholas Tarrier (nicholas.tarrier@manchester.ac.uk)
Graham Dunn (not@valid.com)
Shon Lewis (shon.lewis@manchester.ac.uk)
Til Wykes (til.wykes@kcl.ac.uk)
Linda Davies (linda.m.davies@manchester.ac.uk)
Helen Lester (H.e.lester@bham.ac.uk)
Maria Michail (m.michail@bham.ac.uk)

Version: 5 Date: 26 August 2011

Author's response to reviews:

Dear Editor,

We have amended the contact details of one of the authors in the manuscript.

Kind regards,

Prof. Max Birchwood